Author’s response to reviews

Title: Managing Injuries of the Neck Trial (MINT): design of a randomised controlled trial of treatments for whiplash associated disorders

Authors:

Sarah E Lamb (sallie.lamb@orh.nhs.uk)
Simon Gates (s.gates@warwick.ac.uk)
Martin R Underwood (m.underwood@qmul.ac.uk)
Matthew W Cooke (m.w.cooke@warwick.ac.uk)
Deborah Ashby (d.ashby@wmul.ac.uk)
Ala Szczepura (a.szczepura@warwick.ac.uk)
Mark A Williams (m.a.williams.3@warwick.ac.uk)
Esther M Williamson (e.m.williamson@warwick.ac.uk)
Emma J Withers (emma.withers@warwick.ac.uk)
Shahrul Mt Isa (s.h.mtisa@qmul.ac.uk)
Anil Gumber (a.k.gumber@warwick.ac.uk)

Version: 3 Date: 7 December 2006

Author’s response to reviews:

We have made the requested formatting changes:
1. Trial registration number has been moved to the abstract.
2. The author list is consistent with that in the submission system.
3. Authors: we have adopted the second option, and kept the list of names.
4. Country: we have added the country to the addresses where it was missing.
5. Authors’ contributions. We have added the required sentence.
6. The committee information has been moved to the acknowledgements as requested.